

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Alves, Sandra

Article

The impact of managerial ownership on audit fees: Evidence from Portugal and Spain

Cogent Economics & Finance

Provided in Cooperation with:

Taylor & Francis Group

Suggested Citation: Alves, Sandra (2023) : The impact of managerial ownership on audit fees: Evidence from Portugal and Spain, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 11, Iss. 1, pp. 1-23, https://doi.org/10.1080/23322039.2022.2163078

This Version is available at: https://hdl.handle.net/10419/303932

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU





Cogent Economics & Finance

ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/oaef20

The impact of managerial ownership on audit fees: **Evidence from Portugal and Spain**

Sandra Alves

To cite this article: Sandra Alves (2023) The impact of managerial ownership on audit fees: Evidence from Portugal and Spain, Cogent Economics & Finance, 11:1, 2163078, DOI: 10.1080/23322039.2022.2163078

To link to this article: https://doi.org/10.1080/23322039.2022.2163078

© 2023 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.



6

Published online: 31 Dec 2022.

|--|

Submit your article to this journal 🖸

Article views: 1957



View related articles

View Crossmark data 🗹



Citing articles: 6 View citing articles 🗹



Received: 28 June 2022 Accepted: 21 December 2022

*Corresponding author: Sandra Alves, GOVCOPP, School of Accountancy and Administration, University of Aveiro, Rua Associação Humanitário Bombeiros de Aveiro, Aveiro, Portugal E-mail: sandra.alves@ua.pt

Reviewing editor: David McMillan, University of Stirling, Stirling UK

Additional information is available at the end of the article

FINANCIAL ECONOMICS | RESEARCH ARTICLE

The impact of managerial ownership on audit fees: Evidence from Portugal and Spain

🔆 cogent

economics & finance

Abstract: This study examines the impact of managerial ownership on audit fees in a context of concentrated ownership and poor investor protection. Using samples of Portuguese and Spanish listed companies for the period 2010–2021, the results of this research suggest that there is a non-linear relationship between managerial ownership and audit fee which corresponds to a pattern of "alignmententrenchment-alignment". In Portugal and Spain, the convergence of interest effect dominates the entrenchment effect in the low and high ranges of managerial ownership, leading to the conclusion that a policy of providing management with amounts of equity within these ranges of managerial ownership should reduce agency costs and decrease audit fees. In contrast, in Spain, the entrenchment effect dominates the convergence of interest effect in the intermediate range of managerial ownership, which have the effect of increasing audit fees.

Subjects: Economics; Finance; Business, Management and Accounting

Keywords: Agency theory; managerial ownership; audit fees; alignment of interest hypothesis; entrenchment hypothesis

1. Introduction

The ownership structure is one of the most recognised forms of corporate governance. In particular, managerial ownership is considered by many as an important internal mechanism governance (Brickley et al., 1988; Dixon et al., 2017). Jensen and Meckling (1976) propose that managerial ownership can help to control agency problems since share ownership provides managers with incentives to reduce private perquisite consumption. Managerial shareholdings encourage managers to improve firm value, since managers bear a proportion of the wealth effects as a shareholder. Therefore, shareholdings held by managers help align their interests with those of shareholders (Jensen & Meckling, 1976), and thus reduces the level of opportunistic behaviour of managers (alignment effect) (Velury & Jenkins, 2006; Warfield et al., 1995). Nevertheless, greater managerial ownership can entitle managers to extract rents from shareholders (entrenchment effect). Thus, too much ownership can induce managers to involve with more opportunistic activities that favour their welfare (Cohen et al., 2008; Darrough et al., 1998).

Managers have wide-ranging authority over the financial accountability and reporting process. Consequently, a key element of the financial reporting process is to guarantee an independent verification of the financial statements prepared by the firm's management. Auditing mitigates agency problems by ensuring the quality of the financial statements, which reduces the manager's ability to manipulate accounting information and thus his or her power to extract rent from shareholders (Chan et al., 1993; Desender et al., 2013; Francis & Wang, 2008). Hence, the financial reporting risk and the risk of managers rent extraction can influence the demand for audit services, and thus audit fees (Duellman et al., 2015; Hope et al., 2012; Whisenant et al., 2003). Given that





managerial ownership influences the quality of the firm's financial reporting, it is possible that managerial ownership affects auditor effort and audit fees. In this sense, Carcello et al. (2011) emphasize the need to incorporate management characteristics as a determinant of audit fees due to management's significant influence on accounting, auditing, and internal control.

Prior literature provides evidence to support the relationship between managerial ownership and audit fees. Some studies find a linear relationship (Gotti et al., 2012; Mitra et al., 2007; Park, 2019) while some find a non-linear relationship (Gotti et al., 2012; Shan et al., 2019). The various forms of the relationship between managerial ownership and audit fees found in the above studies all support the "convergence of interest" hypothesis and/or the "entrenchment of interest" hypothesis. Thus, the results of these studies suggest that the impact of management's equity ownership on audit fees varies, depending upon the level of ownership.

However, the existing studies have mostly been carried out in the UK or the US contexts. No research has been done on this topic using data from Portuguese and Spanish corporations. The ownership structure in Portugal and Spain is characterized by ownership concentration, in contrast to the more diffuse form of ownership structure prevalent in the UK and the US (Claessens et al., 2000; Shleifer & Vishny, 1997). In addition, Portugal and Spain are also characterized by poor investor protection. Consequently, the empirical results based on UK and US data may not be applicable to the Portuguese and Spanish context (Fan & Wong, 2002). This motivates our investigation.

Thus, the empirical study in this paper fills a gap in the academic literature by determining the nature of the relationship between managerial ownership and audit fee under the concentrated ownership structure in Portugal and Spain and poor investor protection. The academic literature suggests that the relationship between managerial ownership and audit fees can take a number of forms. Hence, given that no study has been conducted in Portugal and Spain before, we test three potential hypotheses which might explain the nature of the relationship. As a result, the main issues are therefore: is there any relationship between managerial ownership and audit fees in Portugal and Spain? If there is such a relationship, is it a linear one, or is it non-linear in nature? If it is non-linear in nature, does it take the form of a curvilinear (quadratic form) relationship? Or, is it of a non-linear form with a pattern "alignment-entrenchment-alignment" (or "entrenchment-alignment-entrenchment")? Even if it is non-linear in nature, is the pattern of the relationship the same or different from the findings in the US and UK studies? An empirical study based on a sample of non-financial listed Portuguese and Spanish firms-year from 2010 to 2021 is conducted for the present study to answer the above questions.

This paper contributes to the literature in various ways. Firstly, as aforementioned, no research has been done on the relationship between management ownership and audit fees in Portuguese and Spanish listed firms. This paper fills this gap in the academic literature. Secondly, the ownership structure in Portugal and Spain is characterized by ownership concentration, which is in contrast to the more diffuse form of ownership in the US and the UK (Claessens et al., 2000; Shleifer & Vishny, 1997). Therefore, the empirical results based on US and UK data may not be applicable to the Portuguese and Spanish context (Fan & Wong, 2002). In fact, it is doubtful whether the same sort of relationship between managerial ownership and audit fees exists under a concentrated ownership environment. This study therefore examines the impact of managerial ownership on audit fees in a distinctive context and thereby enrich the literature on this issue. Thirdly, Portugal and Spain also present an interesting context to study because they are two countries with weak investor protection (La Porta et al., 1998). Fourthly, the US and UK auditors when compared to other countries' auditors faced significantly higher litigation risk (Chung et al., 2005; Francis, 2006). Therefore, since Portugal and Spain have a lower litigation risk environment (Beveridge et al., 2018), it is possible that managerial ownership would have little impact on audit fees. That is, whether or not managerial ownership affects audit fees in environments with lower litigation risk and lower demand for accounting quality—in code-law countries

such as Portugal and Spain—remains an open question. Hence, this study contributes to literature by responding to calls to examine if and how the relationship between managerial ownership and audit fees holds across jurisdictions (Gotti et al., 2012; Lennox, 2005; Lin & Liu, 2013), mainly in civil law jurisdictions (Shan et al., 2019). Fourth, the findings of this study can provide useful information for auditors, regulators, standard setters and shareholders, mainly whether managerial ownership affects audit fees, especially in countries with a lower litigation risk environment, which is likely to eliminate the deep pockets incentive for investors, and in firms with highly concentrated equity ownership. Finally, findings based on Portuguese and Spanish data also help build a more expansive international understanding of the effect of managerial ownership on audit fees.

This paper is structured as follows. Section two presents theoretical framework and develops testable hypotheses. Section three describes the sample and research design. Section four presents the regression models used in this study. Section five discusses the empirical results. Finally, section six summarises and concludes this study.

2. Theoretical Framework and testable hypotheses

This section first discusses the theoretical framework about the mechanism of managerial shareholdings and agency costs. Next, it discusses the relation between managerial ownership and audit fee. Finally, it develops predictions about the association between management ownership and audit fee.

2.1. Managerial ownership and agency costs

Agency problems arise because "the decision managers who initiate and implement important decisions are not the major residual claimants and therefore do not bear a major share of the wealth effects of their decisions" (Fama & Jensen, 1983, p. 304). Hence, agency costs can be reduced by increasing the level of managers' stock ownership, which may permit a better alignment of their interests with those of shareholders. As managerial ownership increases, managers bear a large fraction of the costs of shirking, perquisite consumption and other value-destroying actions. Additionally, larger managerial ownership reduces the problem of different horizons between owners and managers if share prices adjust rapidly to changes in firm's intrinsic value.

Therefore, managerial ownership is considered an important internal mechanism in the good corporate governance (Jensen & Meckling, 1976). Managerial ownership is expected to reduce the level of agency conflicts because managers bear a proportion of the wealth effects as a shareholder and bear all the costs/benefits associated with the losses/gains in the value of his/ her non-diversified human capital (Fama, 1980). Agency theory predicts that at low levels of managerial equity ownership, managers enjoy personal gain from engaging in suboptimal behaviours. However, as managerial ownership stakes increase managers are less likely to engage in opportunistic behaviour (Jelinek & Stuerke, 2009; Jensen & Meckling, 1976). Therefore, the greater the percentage of shares owned by managers; the more likely managers will make decisions consistent with maximizing shareholders' wealth (the alignment of interest hypothesis).

Contrasting with this positive point of view, an increasing number of authors suggest that managerial holdings may lead to increasing opportunism by managers. In fact, as point out by Demsetz (1983) and Fama and Jensen (1983), managers holding a substantial portion of a company's equity may have enough voting power to ensure that their position inside the firm is secure. High levels of ownership may lead to increasing opportunism by managers, which can cause them to become entrenched within the firm (the entrenchment hypothesis), and consequently increase agency costs (Fama & Jensen, 1983; McConnell & Servaes, 1990; Morck et al., 1988; Short & Keasey, 1999). Entrenched managers are more likely to consume more perquisites and/or reduce the firm's risk profile to protect their own interests (Shleifer & Vishny, 1989). This is because, the higher the ownership stake of the manager, the more difficult it is for outside owners to control the management. At some point, management entrenchment occurs (McConnell & Servaes, 1990; Morck et al., 1988). Morck et al. (1988) find a U shape relationship between

managers' alignment and managers' equity holdings. They suggest the existence of managers' entrenchment for stockholdings between 5% and 25%, and convergence of interests below and above those thresholds. Short and Keasey (1999) also find evidence in accordance with entrenchment when analysing the relationship between performance and managerial ownership. They document a positive association at low levels of ownership, followed by a negative relation (the entrenchment region of managerial ownership occurs between 12% and 40%), and a positive one again at higher levels of management ownership. This is consistent with an alignment of interests between managers and shareholders being dominant at the lowest and highest levels of insider ownership, while entrenchment effects become preponderant in the middle range.

2.2. Managerial ownership and audit fee

Audit fee is the price of audit services provided by external auditors. Audit pricing theory suggests that in a competitive audit market, audit fee is a function of audit effort and the auditor's client-specific business risk (Simunic, 1980; Simunic & Stein, 1996). Financial reporting risk is considered one of the most important risk factors that affect audit pricing (Duellman et al., 2015). Auditors can reduce the risk of accounting misstatements or irregularities by increasing effort, which increases audit quality and audit fees. Hence, the higher the audit risk associated with a particular client firm the more time and effort the audit firm is likely to devote to the audit of the client firm's financial statements (Chen et al., 2016; Ghosh & Tang, 2015; Gul et al., 2003). For example, Bedard and Johbstone (2004) find that auditors increased effort and audit fees for clients with earnings manipulation risk. Therefore, high-risk audits result in higher fees by requiring more audit procedures. Managers have wide-ranging authority over the financial accountability and reporting process. Consequently, managerial ownership could impact the financial reporting risk assessment of the auditor as managerial ownership may lead to reduce/increase the opportunistic behaviours of management.

Previous literature postulates two perspectives to explain the relationship between managerial ownership and audit fees: the demand-side and the supply-side perspectives. Agency theory suggests that client demand for audit quality arises from the incentives to reduce agency costs faced by the firm (Jensen & Meckling, 1976). Thus, the demand-side perspective suggests that the demand for high-quality audit services is a function of information asymmetry and interest conflicts between investors and managers (Healy & Palepu, 2001). Lower agency conflicts decrease the demand for higher audit quality, and hence lower audit fees. In addition, the higher the managerial ownership, the lower the demand for assurance because owners are more actively engaged in day-to-day operations (Niemi, 2005).

Supply-side perspective posits that high-risk audits result in higher fees by requiring more audit procedures. Auditors are supposed to consider the potential impact of managerial ownership on the overall financial reporting process and assess their audit risk accordingly. Such risk assessment is likely to influence their planning of audit efforts as well as the audit fees they charge to their clients. Thus, in higher agency cost settings, auditors are more likely to supply greater effort to prevent misstatement associated with moral hazard and adverse selection problems.

2.3. The predicted association between management ownership and audit fees

The alignment of interest hypothesis suggests a lower demand for and supply of audit services because the conflicts of interest between shareholders and managers are less likely to occur when managers own more company shares. Managerial ownership may reduce opportunistic behaviour among managers when their shareholding increases. As a result, managerial ownership reduces agency costs and the inherent risk of material misstatement. This, lead to less demand of audit services and consequently less audit fees. From the supply-side perspective, the alignment of interest hypothesis suggests a lower earnings manipulation risk, leading to reduced audit risk and less audit work. Consequently, auditors are expected to charge lower audit fees as managerial shareholdings increase. Consistent with the alignment of interest hypothesis, for example, O' Sullivan (2000), using UK data, finds that companies with higher levels of executive share

ownership pay lower audit fees. Gotti et al. (2012) and Mitra et al. (2007), using US data, find that managerial stock ownership is negatively related to audit fees. For a sample of Bangladeshi firms, Karim and Hasan (2012) find that managerial ownership is significantly inversely related to audit fees. Hope et al. (2012), using Norwegian data, find that as managerial ownership increases, audit fees decreases. Barroso et al. (2018), using a sample of 19 countries, find that managerial ownership has a significantly negative relation with audit fees. Park (2019), using data from Korea, also finds a negative association between managerial ownership and audit fees.

The entrenchment hypothesis suggests higher demand and higher supply of assurance services in managerial ownership firms. Firms with higher agency cost (entrenched managers) likely have a greater demand to choose a higher-quality auditor to signal the credibility and transparency of their financial statements and will therefore incur high audit fees (Al-Okaily, 2020; Datar et al., 1991). Auditors are likely to supply higher levels of audit services to detect earnings management and consumption of private benefits at the expense of other shareholders (Al-Okaily, 2020). Consequently, auditors are expected to charge higher audit fees as managerial shareholdings increases. Consistent with the entrenchment hypothesis, Yin and Hung (2011), using Hong Kong data, find that managerial ownership has a significantly positive relationship with audit fees. Oktorina and Wedari (2015), using a sample of Indonesian listed firms, find that as managerial ownership leads to higher audit fees. Alqatamin and Ezeani (2020), using a sample of Jordanian companies, also find a positive relationship between managerial ownership and audit fees.

As referred previously, while the ownership of US and UK listed firms is characterized by dispersed ownership, the ownership of Portuguese and Spain listed firms is characterized by concentrated ownership. In addition, most Portuguese and Spain listed firms are owned and controlled by large shareholders. Thus, we anticipate that the entrenchment effect may not take place (or it may be relatively insignificant) when managerial ownership concentration increases. Instead, the alignment effect may be further reinforced since the managers, who are also one of the large shareholders, are more concerned about the economic value of the firm as their levels of ownership in the firm are higher. Thus, high managerial ownership is likely to reduce the inherent risk of material misstatements in financial reporting. Therefore, this low-risk situation leads to lower engagement effort from auditors and a lower risk premium, hence audit fees should decrease in firms with higher managerial ownership. Moreover, Portugal and Spain have a lower litigation risk environment (Beveridge et al., 2018). Therefore, the management and the audit firms operate in a low litigation risk setting. As a result, the low investor protection in Portugal and Spain also provides no incentives to the auditor to incorporate any risk premium. So, we expect that audit fees decreases as managerial ownership increases. Therefore, we test the following hypothesis:

Alternative Hypothesis (H1): Ceteris paribus, managerial ownership in the firm is likely to be negatively related to audit fees.

However, as argued by LaPorta et al. (1999), there are central agency problems associated with high ownership concentration. The concentration of large shareholders also creates agency problem, namely agency conflict between majority and minority shareholders (Faisal et al., 2020; Gonzalez et al., 2017; Iturriaga & Crisostomo, 2010; Shleifer & Vishny, 1997). There is a tendency for the controlling shareholders to expropriate the minority shareholders when there is high ownership concentration. As referred previously, the ownership of Portuguese and Spain listed firms is characterized by concentrated ownership. Consequently, there is a possibility that at high levels of managerial ownership, the managers are exploiting their powers to the detriment of the minority shareholders. In fact, the expropriation risk is higher when the controlling shareholder is

the manager since the private benefits remain within the manager. When this expropriation happens, the entrenchment effect might override the alignment effect.

Additional audit efforts would be needed to protect the interests of minority shareholders and reduce audit risk. Hence, the resultant effect of "entrenchment of interests" will lead auditors to increase audit work, which increases audit fees. In fact, auditors are likely to supply higher levels of audit services to detect earnings management and consumption of private benefits at the expense of other shareholders. Therefore, it is expected that audit fees will increase at high levels of ownership due to the entrenchment effect. If this is true, the relationship between managerial ownership and audit fees will be curvilinear (quadratic form) (alignment followed by entrenchment). For example, Gotti et al. (2012), using US data, find a curvilinear quadratic relationship ("alignment-entrenchment") between managerial ownership and audit fees.

So, we test the following hypothesis:

Alternative Hypothesis (H2): Ceteris paribus, there is a curvilinear quadratic relationship between managerial ownership and audit fees that is initially negative but eventually increases as managerial ownership increases.

Some studies also suggest that the relationship between managerial ownership and audit fees may be non-linear with a pattern "alignment-entrenchment-alignment" (Gotti et al., 2012; Shan et al., 2019). That is, at low and high levels, managerial ownership can help align the interests of managers with those of shareholders (alignment effect). In contrast, at middle ownership levels, managers may start to behave against the interests of the shareholders (entrenchment effect). For example, Gotti et al. (2012), for a sample of US firms, find a negative effect at low (below 5%) and high (above 25%) levels of managerial ownership on audit fees. Additionally, find a positive relationship between managerial ownership and audit fees at middle ownership levels (between 5% and 25%). Lin and Liu (2013), using a sample of Hong Kong listed firms, find that managerial ownership is negatively associated with audit fees in the low and high regions of managerial ownership, whereas the association is positive in the intermediate region. Shan et al. (2019), for a sample of Australian-listed companies, find that the relationship between managerial ownership and audit fees is positive for companies where managerial ownership is located in the entrenchment region of ownership (between 20% and 55%) but it becomes negative for companies where management ownership is located in the convergence-of-interests regions of ownership (below 20%; above 55%).

At low levels of managerial ownership, there is a possibility that managers will align their interests with the companies (alignment of interest). In contrast, there is also a likelihood that managers will look out for their own interests (entrenchment effect) (Jelinek & Stuerke, 2009; Jensen & Meckling, 1976). Hence, at low levels of managerial ownership, the relationship between managerial ownership and audit fees may be negative (alignment) or positive (entrenchment). However, as managers' incentives to act opportunistically decrease with an increase in managerial ownership (Lin & Liu, 2013), we expect that audit fees will decrease as managerial shareholdings increase in the low region of managerial ownership. In addition, when ownership is concentrated (high non-manager blockholder ownership levels), at low levels of managerial ownership, management faces additional monitoring by blockholders (Jaggi et al., 2009; Jensen & Meckling, 1976). Previous studies suggest that blockholders' monitoring of management's activities is likely to constrain misreporting and earnings management (Bos & Donker, 2004; Chung et al., 2002; Shleifer & Vishny, 1986; Velury & Jenkins, 2006; Xu et al., 2013; Zhong et al., 2007), which may reduce auditors' work, due to reduced audit risk, which is likely to reduce audit fees (Adelopo et al., 2012). In this sense, for example, AlQadasi and Abidin (2018)) find that firms with a higher concentration of ownership are less likely to demand extensive auditing. Therefore, collectively,

in the low region of managerial ownership, we expect a negative relationship between audit fees and managerial ownership.

As non-manager blockholders decrease their shareholdings, they are less likely to constrain misreporting and earnings management. Therefore, at middle ownership levels, managers have stronger control over firms which can enable them to pursue private interests at the expense of other shareholders. It is possibly that managers with significant shareholdings engage in non-value maximizing decisions and opportunistic behaviors to serve their own interests since they are less likely to be disciplined (Lin & Liu, 2013; Shuto & Takada, 2010). As a result, when the risk of entrenchment rises, the need of high audit quality should increase too. In such a situation, the auditor perceives a high level of audit risk, leading to a higher price premium and/or audit efforts and hence higher audit fees. Thus, at middle levels of managerial ownership, we expect a positive relationship between audit fees and managerial ownership. That is, in the intermediate region of managerial shareholding, we expect that the alignment of interests plays an insignificant role, and the effect entrenchment dominates.

At high levels of ownership, managers are more concerned about the wealth consequences of their decisions and, therefore, engage in activities that enhance the firm value. This alignment effect reduces the level of opportunistic behaviour of managers; accordingly, the other share-holders are more likely free of concerns about managerial expropriation. This has the effect of diminishing the inherent risk of material misstatements and therefore the overall audit risk and audit work (Mitra et al., 2007). Therefore, at high managerial ownership, auditors are expected to charge lower audit fees. Niemi (2005) in Finland documents that audit fees are lower for companies majority-owned by their management. Consequently, we expect that at high managerial ownership, the entrenchment effect performances an insignificant role and the alignment of interest effect dominates.

Accordingly, we hypothesise as below:

Alternative Hypothesis (H3a): Ceteris paribus, low levels of managerial ownership in the firm are negatively related to audit fees.

Alternative Hypothesis (H3b): Ceteris paribus, intermediate levels of managerial ownership in the firm are positively related to audit fees.

Alternative Hypothesis (H3c): Ceteris paribus, high levels of managerial ownership in the firm are negatively related to audit fees.

3. Sample and research design

3.1. Sample selection

Our sample includes all the non-financial listed firms of Euronext Lisbon and the Madrid Stock Exchange for the period 2010–2021.

Table 1 details how the selection criteria resulted in a final total unbalanced panel of 1.352 firmyear observations over the 2010 to 2021 period.

The data used in this paper come from the following sources. The Amadeus, a database managed by Bureau Van Dijk and Informa DandB, S.A., the Portuguese Securities Market Supervisory Authority [Comissão de Mercado de Valores Mobiliários (CMVM)] and the Spanish Securities Market Supervisory Authority [Comisión Nacional del Mercado de Valores (CNVM)], which provide the accounting information from annual accounts.

Table 1. Sample selection	criteria during the y	/ears 2010–2021	
		Number of firm years	
Sample selection	Portugal	Spain	Total
Non-financial firms listed	598	1.333	1.931
(-) Football club companies	(45)	-	(45)
(-) Firms with missing data	(141)	(393)	(534)
Number of firm-year observations in the final sample	412	940	1.352

3.2. Research design

3.2.1. Measuring audit fees

Following most audit fee studies (Alkebsee et al., 2021; Barroso et al., 2018; Farooq et al., 2018; Ghafran & O'Sullivan, 2017; Stanley, 2011), the audit fees (Audit_Fee) are measured by the natural log of audit fees paid by the company for audit services during the year.

3.2.2. Measuring managerial ownership

The managerial ownership (MANAG_OWN) is measured as the percentage of shares owned by the manager. As in Gotti et al. (2012), to test the potential curvilinear quadratic relationship between managerial ownership and audit fees, we add the squared term of managerial ownership (MANAG_OWN²). Finally, as in Gotti et al. (2012), Lin and Liu (2013), and Shan et al. (2019), to address the potential a non-linear form with a pattern "alignment-entrenchment-alignment", the total managerial ownership is split into three groups: MANAG_OWN_Low, MANAG_OWN_Medium and MANAG_OWN_High. MANAG_OWN_Low equals MANAG_OWN if 0.00 < MANAG_OWN < 0.05, and 0.05 if MANAG_OWN \geq 0.05; MANAG_OWN_Medium equals MANAG_OWN—0.05 if 0.05 < MANAG_OWN < 0.25, 0.00 if MANAG_OWN \leq 0.05, and 0.20 if MANAG_OWN \geq 0.05; MANAG_OWN \leq 0.05, and 0.20 if MANAG_OWN \geq 0.00 if MANAG_OWN—0.25 if 0.25 < MANAG_OWN < 1.00, and 0.00 if MANAG_OWN \leq 0.25.

3.2.3. Control variables

Given that the managerial ownership is not the only factor affecting audit fees, several control variables are introduced to isolate other factors that may influence the audit fees. Previous research suggests that audit risk, Big 4, free cash flow, board size, ROA, leverage and firm size are associated with audit fees (Al-Okaily, 2020b; Chung et al., 2005; Fleischer & Goettscha, 2012; Francis, 2004; Griffin et al., 2010; Gul & Tsui, 1998; Gull et al., 2021; Habib et al., 2018; Lennox, 2005; Mohammadi et al., 2018; Shailer et al., 2004; Shan et al., 2019, 2020; Simunic, 1980).

Audit risk (Aud_Risk). Audit risk is considered as an important determinant of audit fees (Chan et al., 1993). Certain assets are perceived as being riskier to audit, resulting in higher audit fees. Previous studies suggest that larger amount of inventory and receivables is a signal for higher audit risk (Gandía & Huguet, 2019; Habib et al., 2018; Stanley, 2011). Audit risk is calculated as the sum of inventories and accounts receivables divided by total assets.

Free cash flow (FCF). Jensen (1986) asserts that FCF creates agency problems because of the increased likelihood of value destroying investments. Chung et al. (2005) find that managers of firms with high FCF have the incentives to camouflage their activities by increasing reported earnings through income-increasing discretionary accruals. Consequently, auditors may react by judging firms with high FCF as having higher probability of misstatements and require greater effort. As a result, auditors would impose higher level of audit fees. Griffin et al. (2010) and Gul and

Tsui (1998) find that high FCF companies have higher audit fees. *FCF* is the ratio between the operating cash flows and the total assets.

Board size (*B_Size*). According to Jensen (1993) board size is related to board effectiveness. A higher the number of members on the board lead to a greater the monitoring activity of management. Larger boards seek better monitoring from external auditors. Al-Okaily (2020b), Gull et al. (2021) and Jizi and Nehme (2018) find that firms that have larger boards pay higher audit. *B_Size* is the number of members of the board.

Big 4. Big audit firms earn higher audit fees because of their higher degrees of perceived quality (DeAngelo, 1981). Prior research documents that Big audit firms charge high audit fees (Fleischer & Goettscha, 2012; Francis, 2004; Gandía & Huguet, 2019; Mohammadi et al., 2018; Shailer et al., 2004; Simunic, 1980; Velury & Jenkins, 2006). *Big* 4 takes value 1 if the auditor is a Big 4 and 0 otherwise.

ROA. High profitability firms tend to pay more audit fees to their auditors (Simunic, 1980). Highly profitable firms pay more fees because higher profits may require rigorous auditing testing of the validity for the recognition of revenue and expenses which requires more audit time (Joshi & AL-Bastaki, 2000). Alqatamin and Ezeani (2020) and Yin and Hung (2011) find a positive relationship between profitability and audit fees. *ROA* is the net income deflated by total assets

Leverage. High levered firms can rise the likelihood of financial distress. This likelihood increases audit risk. Therefore, the higher the level of leverage, the more the audit risk and thus higher audit fees. Al-Najjar (2018), Joshi and AL-Bastaki (2000), and Velury and Jenkins (2006) find a positive relationship between debt ratio and audit fees. *Leverage* is calculated as the ratio between the book value of all liabilities and the total assets.

Firm Size (Size). Larger firms are normally more complex and difficult to control. Therefore, larger firms require more audit effort, resulting in higher audit fees (Palmrose, 1986; Simunic, 1980). Al-Najjar (2018), Chung et al. (2005), Gandía and Huguet (2019) and Mohammadi et al. (2018) find that larger firms pay larger audit fees. *Size* is measured as the natural logarithm of market value of equity of each firm.

Finally, to control for variations over time and across firms, we also include year and firm fixed effects.

4. Regression models

The association between managerial ownership and audit fees is examined by estimating the following ordinary least squares (OLS) regressions:

$$\begin{split} \text{Audit}_\text{Fee}_{it} &= \lambda_0 + \lambda_1(\text{MANAG}_\text{OWN}_{it}) + \lambda_2(\text{Aud}_\text{Risk}_{it}) + \lambda_3(\text{FCF}_{it}) + \lambda_4(\text{B}_\text{Size}_{it}) \\ &+ \lambda_5(\text{Big4}_{it}) + \lambda_6(\text{ROA}_{it}) + \lambda_7(\text{Leverage}_{it}) + \lambda_8(\text{Size}_{it}) + \text{Year}_t + \text{Firm}_j + \epsilon_{it} \end{split}$$

$$\end{split}$$

$$\end{split}$$

$$\begin{aligned} \text{Audit}_{\text{F}\text{F}ee}_{it} &= \lambda_0 + \lambda_1(\text{MANAG}_{\text{O}}\text{OWN}_{it}) + \lambda_2(\text{MANAG}_{\text{O}}\text{OWN}_{it}^2)\lambda_3(\text{Aud}_{\text{R}isk}_{it}) \\ &+ \lambda_4(\text{FCF}_{it}) + \lambda_5(\text{B}_{\text{S}ize}_{it}) + \lambda_6(\text{Big}4_{it}) + \lambda_7(\text{ROA}_{it}) + \lambda_8(\text{Leverage}_{it}) \\ &+ \lambda_9(\text{Size}_{it}) + \text{Year}_t + \text{Firm}_i + \epsilon_{it} \end{aligned}$$
(2)

Table 2. Summ	nary of descripti	ve statistics			
	Mean	Median	Min.	Max.	Std. deviation
Panel A—Portugo	II: Number of obser	vations: 412			·
Audit_Fee	5.606	5.672	1.593	9.032	1.025
MANAG_OWN	0.055	0.003	0.000	0.810	0.962
Aud_Risk	0.054	0.008	0.000	0.714	0.809
FCF	0.032	0.031	-2.997	1.544	0.937
B_Size	8.720	9.000	2.000	24.000	3.913
Big4	0.732	1.000	0.000	1.000	0.449
ROA	2.612	2.207	-56.188	56.412	3.961
Leverage	0.482	0.477	0.001	2.604	0.328
Size	19.103	18.989	11.918	23.598	2.346
Panel B—Spain: N	Number of observat	ions: 940			
Audit_Fee	5.903	5.633	1.386	11.002	1.823
MANAG_OWN	0.078	0.004	0.000	0.857	0.897
Aud_Risk	0.183	0.159	0.000	0.701	0.622
FCF	0.080	0.067	-1.086	8.225	1.113
B_Size	12.875	13.000	4.000	32.000	4.605
Big4	0.817	1.000	0.000	1.000	0.392
ROA	3.471	3.055	-58.025	91.200	5.547
Leverage	0.592	0.601	0.003	3.789	0.302
Size	21.103	20.801	16.015	25.937	2.439

Audit_Fee represents the natural log of audit fees paid by the firm for audit services during the year; MANAG_OWN is the percentage of shares owned by the manager; Aud_Risk is the sum of inventories and accounts receivables divided by total assets; FCF is the ratio between the operating cash flows and the total assets; B_Size is the number of members of the board; Big4 dummy variable which takes a value 1 if firm is audited by a Big 4 audit firm and 0 otherwise; ROA is the net income deflated by total assets; Leverage represents the ratio between the book value of all liabilities and the total assets; Size represents the firm's size.

$$\begin{split} \text{Audit}_\text{Fee}_{it} &= \lambda_0 + \lambda_1(\text{MANAG}_\text{OWN}_\text{Low}_{it}) + \lambda_2(\text{MANAG}_\text{OWN}_\text{Medium}_{it}) \\ &+ \lambda_3(\text{MANAG}_\text{OWN}_\text{High}_{it}) + \lambda_4(\text{Aud}_\text{Risk}_{it}) + \lambda_5(\text{FCF}_{it}) + \lambda_6(\text{B}_\text{Size}_{it}) \\ &+ \lambda_7(\text{Big4}_{it}) + \lambda_8(\text{ROA}_{it}) + \lambda_9(\text{Leverage}_{it}) + \lambda_{10}(\text{Size}_{it}) + \text{Year}_t + \text{Firm}_i + \epsilon_{it} \end{split}$$

5. Results and discussion

5.1. Descriptive statistics and correlations

Table 2 presents the sample descriptive statistics for the variables used in this research. Spearman correlations between the explanatory variables are documented in Table 3. Table 4 presents the variation inflation factors (VIF).

Regarding Portugal, Panel A in Table 2 shows that the mean of audit fee (Audit_Fee) is about EUR 720 million with a minimum of EUR 492 thousand and a maximum of EUR 8.450 million. The mean (median) managerial ownership (MANAG_OWN) is 5.5% (0.3%), with a minimum of 0.0% and a maximum of 81%. The mean (median) audit risk is 5.4% (0.8%), with a minimum of 0.0% and a maximum of 71.4%. FCF variable represents on average 0.032 of the total assets of the company (with a median of 0.032). B_size is comprised of approximately 9 members (with a median of 9 members). Big 4 auditors are used by 73.2% of the sample firms. Panel A in Table 2 also shows that the mean (median) ROA is 2.612 (2.207), with a minimum of -56.188 and a maximum of 56.412. Leverage variable represents on average 0.482 of the total assets of the company (with a median of 0.477). The mean of firm size (Size) is about EUR 1.401 million with a minimum of EUR 300 thousand and a maximum of EUR 16.875 million.

(3)

Table 3. Pea	rson correla	tion coefficient	s matrix					
	Audit_Fee	MANAG_OWN	Aud_Risk	FCF	B_size	ROA	Leverage	Size
Audit_Fee	1.00							
MANAG_OWN	-0.09	1.00						
Aud_Risk	-0.11	-0.41	1.00					
FCF	0.14	0.22	-0.06	1.00				
B_Size	0.21	0.08	0.03	0.11	1.00			
ROA	0.17	0.45	-0.09	0.17	0.21	1.00		
Leverage	0.21	-0.12	0.02	-0.08	-0.07	-0.32	1.00	
Size	0.43	0.23	-0.11	-0.17	0.21	0.26	-0.07	1.00

Audit_Fee represents the natural log of audit fees paid by the firm for audit services during the year; MANAG_OWN is the percentage of shares owned by the manager; Aud_Risk is the sum of inventories and accounts receivables divided by total assets; FCF is the ratio between the operating cash flows and the total assets; B_Size is the number of members of the board; ROA is the net income deflated by total assets; Leverage represents the ratio between the book value of all liabilities and the total assets; Size represents the firm's size.

Regarding Spain, Panel B in Table 2 shows that the mean of audit fee (Audit_Fee) is about EUR 1.802 million with a minimum of EUR 400 thousand and a maximum of EUR 31.522 million. The mean (median) managerial ownership (MANAG_OWN) is 7.8% (0.4%), with a minimum of 0.0% and a maximum of 85.7%. The mean (median) audit risk is 18.3% (15.9%), with a minimum of 0.0% and a maximum of 70.1%. FCF variable represents on average 0.080 of the total assets of the company (with a median of 0.067). B_size is comprised of approximately 13 members (with a median of 13 members). Big 4 auditors are used by 81.7% of the sample firms. Panel B in Table 2 also shows that the mean (median) ROA is 3.471 (3.055), with a minimum of -58.025 and a maximum of 91.200. Leverage variable represents on average 0.592 of the total assets of the company (with a median of 0.601). The mean of firm size (Size) is about EUR 5.021 million with a minimum of EUR 406 thousand and a maximum of EUR 96.527 million.

The results of the Table 3 show that all the correlation coefficients are, in general, low (below the 0.9 threshold) (Tabachnick & Fidell, 2001), suggesting the absence of serious statistical problems related with multicollinearity. In the Table 4, the highest VIF is equal to 2.01 (below the 10 threshold) (Belsley et al., 1980). Therefore, we can deduce the absence of any multicollinearity problems.

5.2. Regression results

Table 5, Table 6, and Table 7 report the results of models 1, 2, and 3, respectively. Model 1 includes the managerial ownership variables (MANAG_OWN) as a main treatment variable. To test the potential curvilinear quadratic relationship, model 2 includes both MANAG_OWN and the squared ownership variable (MANAG_OWN²). Finally, model 3 includes the three different levels of managerial ownership (MANAG_OWN_Low, MANAG_OWN_Medium, and MANAG_OWN_High), allowing for variations in the slope coefficient on managerial ownership depending on the ownership level. All the continuous variables are winsorized at the 1% and 99% levels to avoid the effect of outliers.

To both Portugal and Spain, the results in Table 5 show that managerial ownership is negatively related to audit fees. This result is in line with our hypothesis (H1) and the findings of Barroso et al. (2018), Gotti et al. (2012), Hope et al. (2012), Karim and Hasan (2012), Mitra et al. (2007), O' Sullivan (2000), and Park (2019). Thus, consistent with the alignment of interest hypothesis, our results show that managerial ownership is associated with lower audit fees, which suggests that managerial equity ownership alleviates agency conflicts and opportunistic accounting earnings management, leading to lower demand for external audit efforts to resolve agency conflicts, therefore, lower audit fees. From the supply side perspective, auditors respond to the alignment of interests by reducing the level of risk of financial reporting and subsequently the audit fees. This

Table 4. Variation inflation factors (VIF)	
Varaible	VIF
Audit_Fee	1.45
MANAG_OWN	1.91
Aud_Risk	1.01
FCF	1.03
B_Size	1.92
Big4	1.33
ROA	2.01
Leverage	1.81
Size	1.31

Audit_Fee represents the natural log of audit fees paid by the firm for audit services during the year; MANAG_OWN is the percentage of shares owned by the manager; Aud_Risk is the sum of inventories and accounts receivables divided by total assets; FCF is the ratio between the operating cash flows and the total assets; B_Size is the number of members of the board; Big4 dummy variable which takes a value 1 if firm is audited by a Big 4 audit firm and 0 otherwise; ROA is the net income deflated by total assets; Leverage represents the ratio between the book value of all liabilities and the total assets; Size represents the firm's size.

Table 5. Linear regres	sion results (With MAN	AG_OWN)			
	Portu	gal	Spai	in	Total so	Imple
Dependent variable	Audit	Fee	Audit	Fee	Audit	Fee
Independent variables	Coefficient	t-values	Coefficient	t-values	Coefficient	t-values
Constant	0.017	0.073	-0.083	-0.103	0.077	0.506
MANAG_OWN	-0.229	-2.437**	-0.276	-3.211***	-0.372	-3.723***
Aud_Risk	0.255	1.104	0.288	2.390**	0.401	3.002**
FCF	0.122	1.720	0.135	1.989	0.166	1.787
B_Size	0.152	2.566***	0.223	2.904***	0.352	3.055***
Big4	0.013	1.718*	0.370	2.301**	0.501	4.209***
ROA	0.201	0.910	0.055	1.167*	0.099	0.799
Leverage	0.341	2.037**	0.587	4.553***	0.434	4.088***
Size	0.437	3.108***	0.599	4.721***	0.539	3.988***
Year	Yes	5	Yes	5	Yes	5
Firm	Yes	5	Yes	5	Yes	;
Observations	412	2	940)	1.35	2
Adjusted R ²	29.88		37.98		41.07	7%
F-statistic	26.87	5***	32.604	/ *** +	44.505	D***

Audit_Fee represents the natural log of audit fees paid by the firm for audit services during the year; MANAG_OWN is the percentage of shares owned by the manager; Aud_Risk is the sum of inventories and accounts receivables divided by total assets; FCF is the ratio between the operating cash flows and the total assets; B_size is the number of members of the board; Big4 dummy variable which takes a value 1 if firm is audited by a Big 4 audit firm and 0 otherwise; ROA is the net income deflated by total assets; Leverage represents the ratio between the book value of all liabilities and the total assets; Size represents the firm's size.

*** Significant at the 1-percent level; ** Significant at the 5-percent level; * Significant at the 10-percent level.

suggests that auditors view managerial ownership as decreasing managerial incentives to misstate the financial statements and thus audit risk. So, in Portugal and Spain audit fees seem to decrease as managerial ownership increases in listed companies.

	Portu	gal	Spai	in	Total so	Imple
Dependent variable	Audit	Fee	Audit	Fee	Audit	Fee
Independent variables	Coefficient	t-values	Coefficient	t-values	Coefficient	t-values
Constant	0.200	1.089	0.155	0.877	0.987	0.466
MANAG_OWN	-0.171	-2.055**	-0.312	-2.437**	-0.419	-3.158***
MANAG_OWN ²	0.004	0.489	0.018	0.802	0.023	1.209
Aud_Risk	0.188	0.502	0.303	2.233**	0.347	2.343**
FCF	0.167	0.496	0.201	0.667	0.198	0.609
B_Size	0.287	2.987***	0.367	3.102***	0.355	3.098***
Big4	0.587	3.763**	0.315	2.407**	0.109	4.098***
ROA	0.188	0.663	0.465	2.433**	0.299	1.969**
Leverage	0.591	4.176***	0.565	3.830***	0.355	4.391***
Size	0.499	3.985***	0.501	4.087***	0.563	4.007***
Year	Yes	5	Yes		Yes	5
Firm	Yes	5	Yes	;	Yes	5
Observations	412	2	940)	1.35	2
Adjusted R ²	30.76		41.90		45.32	2%
F-statistic	21.556	5***	26.085	D***	34.632	1***

Audit_Fee represents the natural log of audit fees paid by the firm for audit services during the year; MANAG_OWN is the percentage of shares owned by the manager; MANAG_OWN² is the square of the percentage of shares owned by the manager; Aud_Risk is the sum of inventories and accounts receivables divided by total assets; FCF is the ratio between the operating cash flows and the total assets; B_Size is the number of members of the board; Big4 dummy variable which takes a value 1 if firm is audited by a Big 4 audit firm and 0 otherwise; ROA is the net income deflated by total assets; Leverage represents the ratio between the book value of all liabilities and the total assets; Size represents the firm"s size.

*** Significant at the 1-percent level; ** Significant at the 5-percent level; * Significant at the 10-percent level.

From the demand side, one alternative explanation of the negative relationship between managerial ownership and audit fees is that managerial ownership firms may demand lower assurance services to reduce auditor scrutiny over consumption of private benefits and aggressive accounting practices (Al-Okaily, 2020; Desender et al., 2013; Duellman et al., 2015).

Another potential explanation for our result is that this study is conducted on the Portuguese and Spanish capital markets, which have a weak legal environment (La Porta et al., 2000). Francis et al. (2003) documented that countries with weaker legal institutions demanded lower quality audits than countries with stronger legal institutions. In addition, Portuguese and Spanish listed firms have high ownership concentration. This ownership structure can result in lower demand for accounting information (Ball et al., 2000) and external audits (DeFond et al., 2000; Francis et al., 2003).

On the supply side, one alternative explanation of the negative relationship between managerial ownership and audit fees is that the weak legal environment, in Portugal and Spain, lead that the auditor's effort will be more aligned with the shareholders' demands, since auditors are more accommodating to clients' needs as the litigation environment weakens (Hwang & Chang, 2010). The low investor protection in countries such as the two under study provides no incentives to the auditor to incorporate any risk premium (Barroso et al., 2018).

	Portu	gal	Spa	in	Total so	Imple
Dependent variable	Audit	Fee	Audit	Fee	Audit	Fee
Independent variables	Coefficient	t-values	Coefficient	t-values	Coefficient	t-values
Constant	0.045	0.089**	0.178	0.534	0.122	0.378
MANAG_OWN_Low	-0.299	-1.989**	-0.388	-2.087***	-0.403	-3.077***
MANAG_OWN_Medium	0.211	1.082	0.309	2.088**	0.344	2.907***
MANAG_OWN_High	-0.388	-2.899***	-0.491	-2.544***	-0.400	-2.998***
Aud_Risk	0.109	1.463	0.301	2.909**	0.289	2.177**
FCF	0.122	1.309	0.204	1.254	0.178	1.408
B_Size	0.235	2.878***	0.308	2.356***	0.288	2.768***
Big4	0.299	2.099**	0.289	2.078**	0.416	4.798***
ROA	0.099	0.703	0.101	1.998*	0.104	0.698
Leverage	0.409	3.712***	0.489	3.807***	0.413	4.809***
Size	0.411	3.810***	0.459	3.879***	0.409	3.998***
Year	Yes	5	Yes		Yes	
Firm	Yes	5	Yes	;	Yes	i
Observations	412	2	940)	1.35	2
Adjusted R ² F-statistic	32.89		43.97 33.91		53.15 37.387	

Audit_Fee represents the natural log of audit fees paid by the firm for audit services during the year; MANAG_OWN_Low is MANAG_OWN if 0.00 < MANAG_OWN < 0.05, and 0.05 if MANAG_OWN \ge 0.05; MANAG_OWN_Medium is MANAG_OWN—0.05 if 0.05 < MANAG_OWN < 0.25, 0.00 if MANAG_OWN \le 0.05, and 0.20 if MANAG_OWN \ge 0.05; MANAG_OWN—0.05 if 0.05 < MANAG_OWN < 0.25 if 0.25 < MANAG_OWN < 1.00, and 0.00 if MANAG_OWN \ge 0.25; Aud_Risk is the sum of inventories and accounts receivables divided by total assets; FCF is the ratio between the operating cash flows and the total assets; B_Size is the number of members of the board; Big4 dummy variable which takes a value 1 if firm is audited by a Big 4 audit firm and 0 otherwise; ROA is the net income deflated by total assets; Leverage represents the ratio between the book value of all liabilities and the total assets; Size represents the firm's size.

*** Significant at the 1-percent level; ** Significant at the 5-percent level; * Significant at the 10-percent level.

The results of the hypothesis 2 are presented in Table 6. These results show that the coefficient of the variable MANAG_OWN is negative and statistically significant. However, the coefficient of the variable MANAG_OWN² is positive but not statistically significant. Thus, the findings show to both Portugal and Spain, "alignment of interest" at low levels of managerial ownership (MANAG_OWN), but not "entrenchment" at high levels of managerial ownership (MANAG_OWN²). This implies that curvilinear quadratic specification is not suitable for our sample firms. Therefore, the hypothesis 2 is not supported.

Table 7 shows the results of the hypothesis 3. The results indicate that to Spain and the total sample the hypothesis 3 is also supported. The coefficients of the variables MANAG_OWN_Low, MANAG_OWN_Medium and MANAG_OWN_High are all statistically significant. The relationship is found to be of non-linear form. Since the signs of the coefficients of the variables MANAG_OWN_Low, MANAG_OWN_Low, MANAG_OWN_Medium and MANAG_OWN_High are negative, positive and negative respectively, the relationship between audit fees and managerial ownership is "alignment-entrenchment-alignment". To Portugal, the coefficients of the variables MANAG_OWN_Low and MANAG_OWN_High are statistically significant. However, the coefficient of the variable MANAG_OWN_Medium is positive but not statistically significant. Therefore, at low and high levels of managerial ownership, alignment of interests dominates over entrenchment as a determinant

of audit fees. For Portuguese listed firms in the intermediate range of managerial ownership, neither effect dominates.

In general, managerial ownership is beneficial to firm, due to the alignment of interests, as indicated by the significant negative linear relationship under Hypothesis 1. When the percentage of shares owned by managers in a company increases, audit fee decreases. Regarding the non-linear form of the relationship under Hypothesis (3), the results also show, to Portugal and Spain, that at low and high levels of managerial ownership, the relationship between audit fees and managerial ownership is negative, which is consistent with the alignment of interest hypothesis. Thus, this negative relationship suggests a lower demand for audit services when the agency conflicts are lower. On the supply side, this result also suggests that auditors charge lower fees to clients presenting a lower agency conflicts level. The negative relationship at low levels of managerial ownership is also consistent with the conjecture that in ownership concentrated firms, management faces additional monitoring by blockholders. This is expected to constrain misreporting and earnings management, which reduce auditors' work, due to reduced audit risk, which seems to reduce audit fees. Blockholders may have also lower incentives to use auditing services since they bear most of the costs. Thus, blockholders will demand less from the auditor, which will lower the effort and thus the fees (Barroso et al., 2018).

In Spain, at middle levels of managerial ownership, managers appear to be exploiting their power to the detriment of the other shareholders as the relationship between audit fees and managerial ownership becomes positive. Consistent with entrenchment hypothesis, at middle levels of managerial ownership, there seems to be an increase in the minority shareholder expropriation risk and consequently agency conflict, which leads to higher audit fees. Really, from the demand side, non-controlling shareholders (minority shareholders) demand for better audit effort to minimise agency problems and reduce information asymmetry with managers, which results in higher audit fees. From the supply side, the high expropriation risk situation leads to higher engagement effort from auditors, therefore audit fees increase in entrenchedmanagers firms. Hence, our results suggest a lower (higher) demand for audit services when the agency conflicts are lower (higher). The findings also indicate that auditors perceive lower (higher) audit risk when manager interests are (are not) aligned, resulting in lower (higher) audit fees.

The overall findings indicate that the relationship between managerial ownership and audit fee has elements of two functional forms: a negative linear relationship as supported by Hypothesis (1) and a non-linear (with a pattern "alignment-entrenchment-alignment") relationship, to Spain and the total sample, as supported by Hypothesis (3). The non-linear pattern of the relationship is the same findings ("alignment-entrenchment-alignment") of Gotti et al. (2012), Lin and Liu (2013), and Shan et al. (2019). Thus, our results do not support the view that studies based on US and UK evidence may not be necessarily applicable to Portuguese and Spanish listed companies due to the differences in the degree of ownership concentration (Fan & Wong, 2002).

The control variables, board size is positively associated with audit fees, suggesting, as in previous studies, that large boards require more extensive auditing, leading to higher audit fees (Alkebsee et al., 2021; Gull et al., 2021; Jizi & Nehme, 2018). Big 4 is positively and significantly associated with audit fees, suggesting that Big audit firms charge high audit fees (Barroso et al., 2018; Francis, 2004; Mohammadi et al., 2018; Shailer et al., 2004). As in other studies (Habib et al., 2018; Stanley, 2011), to Spain, audit risk has a positive and significant effect on audit fees, which is in line with expectations that higher risk firms pay higher audit fees. To Spain, as in Gandía and Huguet (2019), Ghafran and O'Sullivan (2017), and Joshi and AL-Bastaki (2000), ROA is positively associated with audit fees, suggesting that high profitability firms induce more audit fees. To both Portugal and Spain, the results suggest that larger and higher leveraged firms tend to pay greater

		Port	Portugal			Spain	ii			Total s	Total sample	
	10%-30%	10%-30% threshold	12%-40%	12%-40% threshold	10%-30%	10%-30% threshold	12%-40%	12%-40% threshold	10%-30%	10%-30% threshold	12%-40%	12%-40% threshold
Dependent variable	Audi	Audit Fee	Audi	Audit Fee	Audi	Audit Fee	Audi	Audit Fee	Audi	Audit Fee	Audi	Audit Fee
Independent variables	Coef.	t-values	Coef.	t-values	Coef.	t-values	Coef.	t-values	Coef.	t-values	Coef.	t-values
MANAG_OWN_Low	-0.167	-2.056**	-0.095	-1.986*	-0.207	-2.769**	-0.121	-1.596**	-0.201	-2.023**	-0.197	-2.169**
MANAG_OWN_Medium	660.0	1.809*	0.109	2.007*	0.126	2.318*	0.099	2.001*	0.209	2.760**	0.200	2.718**
MANAG_OWN_High	-0.136	-2.167**	-0.105	-1.809**	-0.222	-2.420**	-0.133	-1.393**	-0.217	-2.655**	-0.223	-2.420**
Control variables	~	Yes	×	Yes	×	Yes	×	Yes	×	Yes	γ.	Yes
Year	×	Yes	×	Yes	×	Yes	×	Yes	×	Yes	γ.	Yes
Firm	~	Yes	×	Yes	×	Yes	×	Yes	×	Yes	γ.	Yes
Observations	4	412	(7	412	6	940	6	940	1.0	1.352	1.3	1.352
Adjusted R ²	41.(41.09%	42.(42.06%	53.(53.00%	55.(55.07%	62.(62.09%	2.69	69.20%
F-statistic	39.6	39.676***	47.9	47.986***	46.0	46.099***	54.6	54.643***	54.6	54.643***	79.8	79.885***

Table 9. Regression Res	sults—Cubic s	specificati	on			
	Portu	gal	Spa	in	Total sa	ımple
Dependent variable	Audit	Fee	Audit	Fee	Audit	Fee
Independent variables	Coefficient	t-values	Coefficient	t-values	Coefficient	t-values
MANAG_OWN	-0.028	-1.699*	-0.205	-1.989**	-0.311	-2.765***
MANAG_OWN ²	0.163x10 ⁻²	0.978	0.189x10 ⁻²	1.897*	0.202x10 ⁻³	2.002**
MANAG_OWN ³	-0.231x10 ⁻⁴	-2.012**	-0.298x10 ⁻⁵	-2.176**	-0.298x10 ⁻⁶	-2.563***
Control variables	Yes	;	Yes		Yes	
Year	Yes		Yes		Yes	
Firm	Yes		Yes		Yes	
Observations	412	2	940)	1.35	2
Adjusted R ² F-statistic	20.72 26.053		29.08 30.746		37.73 34.734	

audit fees consistent with other studies (Fleischer & Goettscha, 2012; Ghafran & O´Sullivan, 2017; Joshi & AL-Bastaki, 2000; Mohammadi et al., 2018; Sellami & Cherif, 2020).

5.3. Supplementary analysis

As observed by Morck et al. (1988), the literature does not define where the intermediate shareholding region lies. Therefore, the ownership thresholds that define the various managerial shareholding regions are "ad hoc". We therefore test alternative thresholds to check whether our findings are robust. Following Lin and Liu (2013) and Short and Keasey (1999), we test also the thresholds to 10%-30% and 12%-40%, respectively. The regression results (Table 8) for the main variables of interest support Hypothesis 3. However, the coefficients for the regions of managerial ownership become much less significant, suggesting that the most appropriate entrenchment region in the Spain and in the total sample is at 5%-25% with turning points at the significance levels of 5% and 1%, respectively. In the model 3, we predict a non-linear relationship between managerial ownership and audit fees, incorporating three different regions of managerial ownership in a piecewise regression. However, the piecewise linear regression analysis assumes sudden rather than gradual turning points at varied managerial ownership levels (Lin & Liu, 2013). Some studies avoid this problem by using additional cubic form in the regression analysis (Lennox, 2005; Lin & Liu, 2013; Shan et al., 2019; Short & Keasey, 1999). Thus, we re-estimate the model 3 including a cubic specification of MANAG OWN to check whether our results are robust. The results (Table 9) show that including a cubic specification of MANAG OWN does not affect main findings. In the models 1, 2 and 3, we assume that managerial ownership is exogenous. However, previous studies suggest that managerial ownership is endogenous (e.g., Demsetz & Villalonga, 2013b; Gugler & Weigand, 2003; Himmelberg et al., 1999; Sellami & Cherif, 2020; Shahveisi et al., 2017; Villalonga, 2019b). Therefore, we investigate whether there is an endogeneity problem for managerial ownership, using the two-stage least squares (2SLS) regression specification. Following the methodology used by Lennox (2005) and Shan et al. (2019, 2020), we estimate an OLS model with managerial ownership as the dependent variable, and use the coefficient estimates to obtain the predicted level of ownership (MANAG OWN Predicted). We then include the MANAG_OWN_Predicted as the measure of MANAG_OWN in the models 1, 2 and 3. The results (Table 10) remain qualitatively similar to those reported in the Tables 5, 6 and 7, suggesting that MANAG_OWN_Predicted is endogenous. Thus, the MANAG_OWN variable is exogenous in our sample. Hence, the 2SLS regression results for the main variables of interest support our Hypotheses 1 and 3. However, for Portuguese listed firms in the intermediate range of managerial ownership, neither effect dominates.

Table 10. 2SLS regressions results	esults								
		Portugal			Spain			Total sample	
Hypotheses	H1	H2	H3	H1	H2	H3	H1	H2	H3
Dependent variable		Audit Fee			Audit Fee			Audit Fee	
	Coef.	Coef.	Coef.	Coef.	Coef.	Coef.	Coef.	Coef.	Coef.
Independent variables	(t-values)	(t-values)	(t-values)	(t-values)	(t-values)	(t-values)	(t-values)	(t-values)	(t-values)
MANAG_OWN_Predicted	-0.189 (-2.262**)	-0.152 (-2.008**)	I	-0.201 (-3.079***)	-0.289 (-2.356**)	1	-0.211 (-3.508***)	-0.298 (-3.673***)	I
MANAG_OWN ² _Predicted	1	0.098 (0.627)	ı	I	0.267 (2.156**)	1	I	0.388 (2.679***)	ı
MANAG_OWN_Low_Predicted	ı	I	-0.201 (-1.798**)	I	1	-0.289 (-2.001**)		1	-0.353 (-3.001***)
MANAG_OWN_Medium_Predicted	ı	I	0.179 (0.989)	I	I	0.291 (1.989**)	I	ı	0.325 (2.799***)
MANAG_OWN_High_Predicted	1	1	-0.278 (-2.687***)	1	ı	0.298 (3.001***)	1	1	-0.307 (-2.760**)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Firm	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations		412			940			1.352	
Adjusted R ²	27.39%	26.97%	28.07%	38.44%	40.92%	41.87%	41.08%	45.06%	53.83%
F-statistic	21.098***	19.576***	24.736***	29.780***	22.908***	30.273***	37.902***	33.987***	38.098***

ults	
ons res	-

6. Conclusion

Using samples of Portuguese and Spanish listed companies for the period 2010–2021, this paper explores the impact of managerial ownership on audit fees. Managerial ownership is considered an effective mechanism to address agency problem (align management with shareholders). Agency theory predicts that when managers hold low levels of equity ownership in a firm, their incentives to shirk and consume perquisites rise. However, as managerial ownership stakes increase, managers are less likely to engage in opportunistic behaviour (Jensen & Meckling, 1976). Audit pricing theory suggests that auditors might charge lower fees to less riskier clients.

Our results show that managerial ownership firms incur lower audit fees, which is driven by firms' lower demand for external auditing services (due to the less severe agency problems) and auditors perceived lower audit risk for managerial ownership firms. The results also show that, to Portugal and Spain, when managerial ownership is low (<5%) and high (>25%), the convergence-of-interest dominates, thereby resulting in a negative relation between managerial ownership and audit fees. In Spain, the entrenchment effect dominates the alignment of interest effect in the intermediate (5%-25%) range of managerial ownership, which have the effect of increasing audit fees. For the total sample and Spain, the relationship between managerial ownership and audit fees has elements of the two functional forms: a negative linear form and a non-linear form ("alignment-entrenchment-alignment").

Hence, consistent with prior literature, this study indicates that in Portuguese and Spanish listed firms the alignment effect is very strong, as managerial ownership increases (a negative linear relationship). There are also alignment effects at both ends of the managerial ownership levels (negative relationships at low and high levels of managerial ownership). This suggests that managerial ownership (low and high levels) reduces agency problems in financial reporting and decrease the risk of accounting misstatements or irregularities, consequently, the scope of audit work becomes lower leading to lower audit fees. At the same time, to Spain and the total sample, at middle levels of managerial ownership the entrenchment effects become dominant, as denoted by the entrenchment ranges in the non-linear relationship.

The results of this study make the following contributions. First, this study contributes to the literature in corporate governance by showing that managerial ownership affects audit fees. Therefore, this paper contributes to understanding how managerial ownership can influence auditors' perceptions of client risk and, in turn, affects audit fees. Second, this study also contributes to the literature by showing that managerial ownership affects audit fees in environments with lower litigation risk in code-law countries such as Portugal and Spain. Third, this paper has also implications for cross-national governance and auditing price research. Fourth, the findings are relevant for countries with an institutional environment similar to that of Portugal and Spain.

Finally, the results of this study are likely of interest to policymakers by offering useful feedback to consider managerial ownership during their rulemaking processes. It also helps financial market participants with making better-informed investment decisions and aids other interested parties in gaining a better understanding of the role played by managerial ownership in audit fees.

This study has certain limitations as well. Firstly, we focus exclusively on the impact of managerial ownership on audit fees. Consequently, this study neglects the impact of other corporate governance mechanisms on audit fees. Secondly, the sample of this study is constituted only of listed companies. Hence, our findings may not be generalized to small and non-listed firms. Lastly, our sample is restricted to Portugal and Spain. So, the evidence provided might not necessarily generalize to other environmental contexts.

For future research, it would be of interest to examine the impact of other governance mechanisms on audit fees. This research might be also extended towards other European countries. Small and mediumsized enterprises (SMEs) are the backbone of Europe's economy. Thus, in this vein, for future research it would be also interesting to replicate the analyses using data from Portuguese/Spanish/European SMEs.

Funding

This work was financially supported by the research unit on Governance, Competitiveness and Public Policy (UIDB/ 04058/2020) + (UIDP/04058/2020), funded by national funds through FCT - Fundação para a Ciência e a Tecnologia;This work was financially supported by the research unit on Governance, Competitiveness and Public Policy (UIDB / 04058/2020) + (UIDP / 04058/2020), funded by national funds through FCT - Fundação para a Ciência e a Tecnologia;

Author details

Sandra Alves¹

E-mail: sandra.alves@ua.pt

¹ GOVCOPP, School of Accountancy and Administration, University of Aveiro, Rua Associação Humanitário Bombeiros de Aveiro, Aveiro, Portugal.

Disclosure statement

No potential conflict of interest was reported by the author(s).

Citation information

Cite this article as: The impact of managerial ownership on audit fees: Evidence from Portugal and Spain, Sandra Alves, *Cogent Economics & Finance* (2023), 11: 2163078.

References

- Adelopo, I., Jallow, K., & Scott, P. (2012b). "Multiple large ownership structure, audit committee activity and audit fees." Journal of Applied Accounting Research 13, no. 2: 100–121. https://doi.org/10.1108/ 09675421211254821
- Alkebsee, R., Tian, G. L., Usman, M., Siddique, M. A., & Alhebry, A. (2021b). Gender diversity in audit committees and audit fees: Evidence from China. *Managerial Auditing Journal*, 36(1), 72–104. https:// doi.org/10.1108/MAJ-06-2019-2326
- Al-Najjar, B. (2018). Corporate governance and audit features: SMEs evidence. Journal of Small Business and Enterprise Development, 25(1), 63–179. https:// doi.org/10.1108/JSBED-08-2017-0243
- Al-Okaily, J. (2020b). "The effect of family control on audit fees during financial crisis." *Managerial Auditing Journal* 35, no. 1: 645–665. https://doi.org/10.1108/ MAJ-12-2018-2114
- AlQadasi, A., & Abidin, S. (2018). The effectiveness of internal corporate governance and audit quality: The role of ownership concentration – Malaysian evidence. *Corporate Governance*, 18(2), 233–253. https://doi.org/10.1108/CG-02-2017-0043
- Alqatamin, R. M., & Ezeani, E. (2020). The impact of fair value estimates on audit fees: Evidence from the financial sector in Jordan. Journal of Accounting in Emerging Economies, 11(2), 176–193. https://doi.org/ 10.1108/JAEE-09-2019-0184
- Ball, R., Kothari, S. P., & Robin, A. (2000). The effect of international institutional factors on properties of accounting earnings. *Journal of Accounting and Economics*, 29(1), 1–51. https://doi.org/10.1016/ S0165-4101(00)00012-4
- Barroso, R., Ali, C. B., & Lesage, C. (2018). "Blockholders" ownership and audit fees: The impact of the corporate governance model. *European Accounting Review*, 27(1), 149–172. https://doi.org/10.1080/09638180. 2016.1243483
- Bedard, J. C., & Johbstone, K. M. (2004). Earnings manipulation risk, corporate governance risk, and auditors planning and pricing decisions. *The Accounting Review*, 79(2), 277–304. https://doi.org/10.2308/accr. 2004.79.2.277

- Belsley, D. A., Kuh, E., & Welsch, R. E. (1980). Regression diagnostics: Identifying influential data and sources of collinearity. John Wiley. Beveridge, F., Nott, and Stephen, K. Making Women Count: Integrating Gender into Law and Policy-Making, Routledge Revivals, London, 2018
- Beveridge, F., Nott, S., & Stephen, K. (2018). Making Women Count: Integrating Gender into Law and Policy-Making. Routledge Revivals.
- Bos, A. D., & Donker, H. (2004). Monitoring accounting changes: Empirical evidence from the Netherlands. *Corporate Governance: An International Review*, 12 (1), 60–73. https://doi.org/10.1111/j.1467-8683.2004. 00343.x
- Brickley, J., Lease, R., & Smith, C. (1988). Ownership structure and voting on antitakeover amendments. *Journal of Financial Economics*, 20(1–2), 267–291. https://doi.org/10.1016/0304-405X(88)90047-5
- Carcello, J. V., Hermanson, D. R., & Ye, Z. S. (2011). Corporate governance research in accounting and auditing: Insights, practice implications, and future research directions. Auditing: A Journal of Practice and Theory, 30(3), 1–31. https://doi.org/10.2308/ajpt-10112
- Chan, P., Ezzamel, M., & Gwilliam, D. (1993). Determinants of audit fees for quoted UK companies. *Journal of Business, Finance and Accounting*, 20(6), 765–785. https://doi.org/10.1111/j.1468-5957.1993.tb00292.x
- Chen, P. F., He, S., Ma, Z., & Stice, D. (2016). The information role of audit opinions in debt contracting. *Journal of Accounting and Economics*, 61(1), 121–144. https://doi.org/10.1016/j.jacceco.2015.04. 002
- Chung, R., Firth, M., & Kim, J. (2002). Institutional monitoring and opportunistic earnings management. *Journal of Corporate Finance*, 8(1), 29–48. https://doi. org/10.1016/S0929-1199(01)00039-6
- Chung, R., Firth, M., & Kim, J. B. (2005). Earnings management, surplus free cash flow, and external monitoring. *Journal of Business Research*, 58(6), 766–776. https://doi.org/10.1016/j.jbusres.2003.12. 002
- Claessens, S., Djankov, S., & Lang, L. H. P. (2000). The separation of ownership and control in East Asian Corporations. Journal of Financial Economics, 58(1–2), 81–112. https://doi.org/10.1016/S0304-405X(00) 00067-2
- Cohen, D. A., Dey, A., & Lys, T. Z. (2008). Real and accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. *The Accounting Review*, 83(3), 757–787. https://doi.org/10.2308/accr.2008.83. 3.757
- Darrough, M. N., Pourjalali, H., & Saudagaran, S. (1998). Earnings management in Japanese companies. *The International Journal of Accounting*, 33(3), 313–334. https://doi.org/10.1016/S0020-7063(98)90034-6
- Datar, S., Feltham, G., & Hughes, J. (1991). The role of audits and audit quality in valuing new issues. Journal of Accounting and Economics, 14(1), 3-49. https://doi.org/10.1016/0167-7187(91)90057-R
- DeAngelo, L. (1981). Auditor size and auditor quality. Journal of Accounting and Economics, 3(3), 183–199. https://doi.org/10.1016/0165-4101(81)90002-1
- DeFond, M. L., Wong, T. J., & Li, S. (2000). The impact of improved auditor Independence on audit market concentration in *China. Journal of Accounting and Economics*, 28(3), 269–305. https://doi.org/10.1016/ S0165-4101(00)00005-7
- Demsetz, H. (1983). The Structure of Ownership and the Theory of the Firm. *Journal of Law and Exonomics*, 26 (2), 375–390.

- Demsetz, H., & Villalonga, B. (2001). Ownership structure and corporate performance. Journal of Corporate Finance, 7(3), 209–233. https://doi.org/10.1016/ S0929-1199(01)00020-7
- Desender, K. A., Aguilera, R. V., Crespi, R., & García-cestona, M. (2013b). "When does ownership matter? Board characteristics and behavior." *Strategic Management Journal* 34, no. 7: 823–842. https://doi.org/10.1002/smj.2046
- Dixon, R., Guariglia, A., & Vijayakumar, R. (2017). Managerial ownership, corporate governance, and firms" exporting decisions: Evidence from Chinese listed companies. *The European Journal of Finance*, 23(7–9), 802–840. https:// doi.org/10.1080/1351847X.2015.1025990
- Duellman, S., Hurwitz, H., & Sun, Y. (2015). Managerial overconfidence and audit fees. Journal of Contemporary Accounting and Economics, 11(2), 148–165. https://doi.org/10.1016/j.jcae.2015.05.001
- Faisal, F., Majid, M. S. A., Sakir, A., & McMillan, D. (2020). Agency conflicts, firm value, and monitoring mechanisms: An empirical evidence from Indonesia. *Cogent Economics and Finance*, 8(1), 1822018. https://doi.org/10.1080/23322039.2020.1822018
- Fama, E. F. (1980). Agency Problems and the Theory of the Firm. Journal of Political Economy, 88(2), 288–307. https://doi.org/10.1086/260866
- Fama, E. F., & Jensen, M. C. (1983). Separation of Ownership and Control. The Journal of Law and Economics, 2(26), 301–325. https://doi.org/10.1086/ 467037
- Fan, J., & Wong, T. J. (2002). Corporate ownership structure and the informativeness of accounting earnings in East Asia. Journal of Accounting and Economics, 33(3), 401–425. https://doi.org/10.1016/S0165-4101(02) 00047-2
- Farooq, M. U., Kazim, I., Usman, M., & Latif, I. (2018). Corporate governance and audit fees: Evidence from a developing country. *Pakistan Journal of Commerce* and Social Sciences, 12(1), 94–110. https://hdl.handle. net/10419/188337
- Fleischer, R., & Goettscha, M. (2012). Size effects and audit pricing: Evidence from Germany. Journal of International Accounting, Auditing and Taxation, 21 (2), 156–168. https://doi.org/10.1016/j.intaccaudtax. 2012.07.005
- Francis, J. R. (2004). The effect of auditor firm size on audit prices: A study of the Australian market. *Journal of Accounting and Economics*, 6(2), 131–151. https://doi.org/10.1016/0165-4101(84)90010-7
- Francis, J. R. (2006). Are auditors compromised by Nonaudit services? Assessing the evidence. *Contemporary Accounting Research*, 23(3), 747–760. https://doi.org/10.1506/4VD9-AE3K-XV7L-XT07
- Francis, J. R., Khurana, I. K., & Pereira, R. (2003). The role of accounting auditing in corporate governance and the development of financial markets around the world. Asia-Pacific Journal of Accounting and Economics, 10(1), 1–30. https://doi.org/10.1080/ 16081625.2003.10510613
- Francis, J. R., & Wang, D. (2008). The joint effect of investor protection and big 4 audits on earnings quality around the world. *Contemporary Accounting Research*, 25(1), 157–191. https://doi.org/10.1506/ car.25.1.6
- Gandía, J. L., & Huguet, D. (2019). Audit fees and cost of debt: Differences in the credibility of voluntary and mandatory audits. *Economic Research-Ekonomska Istraživanja*, 33(1), 3071–3092. https://doi.org/10. 1080/1331677X.2019.1678501
- Ghafran, C., & O'Sullivan, N. (2017). The impact of audit committee expertise on audit quality: Evidence from

UK audit fees. The British Accounting Review, 49(6), 578–593. https://doi.org/10.1016/j.bar.2017.09.008

- Ghosh, A., & Tang, C. Y. (2015). Assessing financial reporting quality of family firms: The auditors perspective. *Journal of Accounting and Economics*, 60 (1), 95–116. https://doi.org/10.1016/j.jacceco.2015. 03.002
- Gonzalez, M., Molina, C. A., Pablo, E., & Rosso, J. W. (2017). The effect of ownership concentration and composition on dividends: Evidence from Latin America. *Emerging Markets Review*, 30, 1–18. https://doi.org/ 10.1016/j.ememar.2016.08.018
- Gotti, G., Han, S., Higgs, J. L., & Kang, T. (2012). Managerial stock ownership, analyst coverage, and audit fee. Journal of Accounting, Auditing and Finance, 27(3), 412–437. https://doi.org/10.1177/ 0148558X11409158
- Griffin, P. A., Lont, D. H., & Sun, Y. (2010). Agency problems and audit fees: Further tests of the free cash flow hypothesis. Accounting and Finance, 50(2), 321–350. https://doi.org/10.1111/j.1467-629X.2009.00327.x
- Gugler, K., & Weigand, J. (2003). Is ownership really endogenous? Applied Economics Letters, 10(8), 483–486. https://doi.org/10.1080/ 1350485032000095357
- Gul, F. A., Chen, C. J. P., & Tsui, J. S. L. (2003). Discretionary accounting accruals, managers incentives, and audit fees. *Contemporary Accounting Research*, 20(3), 441–464. https://doi.org/10.1506/686E-NF2J-73X6-G540
- Gull, A. A., Atif, M., Issa, A., Usman, M., & Siddique, M. A. (2021). Female CEO succession and audit fees: Evidence from China. *Managerial Auditing Journal*, 36(3), 485–509. https://doi.org/10.1108/MAJ-09-2020-2824
- Gul, F. A., & Tsui, J. (1998). A test of the free cash flow and debt monitoring hypotheses: Evidence from audit pricing. Journal of Accounting and Economics, 24(2), 219–237. https://doi.org/10.1016/S0165-4101(98) 00006-8
- Habib, A., Bhuiyan, B. U., & Rahman, A. (2018). "Problem" directors and audit fees. *International Journal of Auditing*, 23(1), 125–143. https://doi.org/10.1111/ijau. 12150
- Healy, P. M., & Palepu, K. G. (2001). Information asymmetry, corporate disclosure, and the capital markets: A review of the empirical disclosure literature. *Journal of Accounting and Economics*, 31(1–3), 405–440. https:// doi.org/10.1016/S0165-4101(01)00018-0
- Himmelberg, C. P., Hubbard, R. G., & Palia, D. (1999). Understanding the determinants of managerial ownership and the link between ownership and performance. *Journal of Financial Economics*, 53(3), 353–384. https:// doi.org/10.1016/S0304-405X(99)00025-2
- Hope, O. K., Langli, J. C., & Thomas, W. B. (2012). Agency conflicts and auditing in private firms. Accounting, Organizations and Society, 37(7), 500–517. https:// doi.org/10.1016/j.aos.2012.06.002
- Hwang, N.-C. R., & Chang, C. J. (2010). Litigation environment and auditors" decisions to accept clients" aggressive reporting. *Journal of Accounting and Public Policy*, 29(3), 281–295. https://doi.org/10.1016/ j.jaccpubpol.2010.02.002
- Iturriaga, F. J. L., & Crisostomo, V. L. (2010). Do leverage, dividend payout, and ownership concentration influence firms" value creation? An analysis of Brazilian firms. *Emerging Market Finance and Trade*, 46(3), 80–94. https://doi.org/10.2753/REE1540-496X460306
- Jaggi, B., Leung, S., & Gul, F. (2009). Family control, board Independence and earnings management: Evidence based on Hong Kong firms. Journal of Accounting and

Public Policy, 28(4), 281–298. https://doi.org/10.1016/ j.jaccpubpol.2009.06.002

- Jelinek, K., & Stuerke, P. S. (2009). The nonlinear relation between agency costs and managerial equity ownership: Evidence of decreasing benefits of increasing ownership. International Journal of Managerial Finance, 5(2), 156–178. https://doi.org/10.1108/ 17439130910947886
- Jensen, M. C. (1986). Agency costs of free cash flow, corporate finance, and takeovers. The American Economic Review, 76(2), 323–329. https://www.jstor. org/stable/1818789
- Jensen, M. C. (1993). The modern industrial revolution, exit, and the failure of internal control systems. *The Journal of Finance*, 3(48), 831–880. https://doi.org/10. 1111/j.1540-6261.1993.tb04022.x
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: Managerial behavior, agency costs, and ownership structure. Journal of Financial Economics, 3(4), 305–360. https://doi.org/10.1016/0304-405X(76) 90026-X
- Jizi, M., & Nehme, R. (2018). Board monitoring and audit fees: The moderating role of CEO/ chair dual roles. Managerial Auditing Journal, 33(2), 217–243. https:// doi.org/10.1108/MAJ-10-2016-1464
- Joshi, P. L., & AL-Bastaki, H. (2000). Determinants of audit fees: Evidence from the Companies Listed in Bahrain. *International Journal of Auditing*, 4(2), 129–138. https://doi.org/10.1111/1099-1123.00308
- Karim, A. W., & Hasan, T. (2012). The market for audit services in Bangladesh. Journal of Accounting in Emerging Economies, 2(1), 50–66. https://doi.org/10. 1108/20421161211196120
- La Porta, R., Lopes-de-Silanes, F., Shleifer, A., & Vishny, R. W. (1998). Law and Finance. Journal of Political Economy, 106(6), 1113–1155. https://doi.org/ 10.1086/250042
- LaPorta, R., Lopez-de-Silanes, F., & Shleifer, A. (1999). Corporate ownership around the world. *Journal of Finance*, 54(2), 471–517. https://doi.org/10.1111/ 0022-1082.00115
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (2000). Investor protection and corporate governance. *Journal of Financial Economics*, 58(1–2), 3–27. https://doi.org/10.1016/S0304-405X(00) 00065-9
- Lennox, C. (2005). Management ownership and audit firm size. Contemporary Accounting Research, 22(1), 205–227. https://doi.org/10.1506/K2CG-U6V0-NPTC-EQBK
- Lin, Z. J., & Liu, M. (2013). The effects of managerial shareholding on audit fees: Evidence from Hong Kong. International Journal of Auditing, 17(3), 227–245. https://doi.org/10.1111/ijau.12000
- McConnell, J. J., & Servaes, H. (1990). Additional evidence on equity ownership and corporate value. Journal of Financial Economics, 27(2), 595–612. https://doi.org/ 10.1016/0304-405X(90)90069-C
- Mitra, S., Hossain, M., & Deis, D. R. (2007). The empirical relationship between ownership characteristics and audit fees. *Review of Quantitative Finance and Accounting*, 28(3), 257–285. https://doi.org/10.1007/ s11156-006-0014-7
- Mohammadi, M. T., Kardan, B., & Salehi, M. (2018). The relationship between cash holdings, investment opportunities and financial constraint with audit fees. *Asian Journal of Accounting Research*, 3(1), 15–27. https://doi.org/10.1108/AJAR-07-2018-0016
- Morck, R., Shleifer, A., & Vishny, R. W. (1988). Management ownership and market valuation - an

empirical analysis. Journal of Financial Economics, 20 (1-2), 293-315. https://doi.org/10.1016/0304-405X (88)90048-7

- Niemi, L. (2005). Audit effort and fees under concentrated client ownership: Evidence from four international audit firms. *International Journal of Accounting*, 40 (4), 303–323. https://doi.org/10.1016/j.intacc.2005. 09.006
- Oktorina, M., & Wedari, L. K. (2015). An empirical investigation on ownership characteristics, activities of the audit committee, and audit fees. *Applied Finance and Accounting*, 1(1), 20–29. https://doi.org/10.11114/afa. v1i1.639
- O' Sullivan, N. (2000). The impact of board composition and ownership on audit quality: Evidence from large UK companies. *British Accounting Review*, 33(4), 397–414. https://doi.org/10.1006/bare.2000.0139
- Palmrose, Z. (1986). Audit fees and auditor size. Journal of Accounting Research, 24(1), 97–110. https://doi.org/ 10.2307/2490806
- Park, B.-J. (2019). Audit committees and managerial influence on audit quality: "Voluntary" versus "Mandatory" approach to corporate governance. *Australian Accounting Review, 29*(88), 266–280. https://doi.org/10.1111/auar.12263
- Sellami, Y. M., & Cherif, I. (2020). Female audit committee directorship and audit fees. *Managerial Auditing Journal*, 35(3), 398–428. https://doi.org/10.1108/MAJ-12-2018-2121
- Shahveisi, F., Khairollahi, F., & Alipour, M. (2017). Does ownership structure matter for corporate intellectual capital performance? An empirical test in the Iranian context. *Eurasian Business Review*, 7(1), 67–91. https://doi.org/10.1007/s40821-016-0050-8
- Shailer, G., Cummings, L., Vatuloka, E., & Welch, S. (2004). Discretionary pricing in a monopolistic audit market. *International Journal of Auditing*, 8(3), 263–277. https://doi.org/10.1111/j.1099-1123.2004. 00095.x
- Shan, Y. G., Troshani, I., & Loftus, J. (2020). Managerial ownership, audit committees and non-audit services. *Australian Journal of Management*, 46(3), 1–18. https://doi.org/10.1177/0312896220945763
- Shan, Y. G., Troshani, I., & Tarca, A. (2019). Managerial ownership, audit firm size, and audit fees: Australian evidence. Journal of International Accounting, Auditing and Taxation, 35, 18–36. https://doi.org/10. 1016/j.intaccaudtax.2019.05.002
- Shleifer, A., & Vishny, R. W. (1986). Large shareholders and corporate control. *Journal of Political Economy*, 94(3), 461–488. https://doi.org/10.1086/261385
- Shleifer, A., & Vishny, R. W. (1989). Managerial Entrenchment: The Case of Manager-Specific Investments. *Journal of Financial Economics*, 5(1), 123–139. https://doi.org/10. 1016/0304-405X(89)90099-8
- Shleifer, A., & Vishny, R. W. (1997). A survey of corporate governance. *Journal of Finance*, 52(2), 737–783. https:// doi.org/10.1111/j.1540-6261.1997.tb04820.x
- Short, H., & Keasey, K. (1999). Managerial ownership and the performance of firms: Evidence from the UK. *Journal of Corporate Finance*, 5(1), 79–101. https:// doi.org/10.1016/S0929-1199(98)00016-9
- Shuto, A., & Takada, T. (2010). Managerial ownership and accounting conservatism in Japan: A test of management. Journal of Business Finance and Accounting, 37(7–8), 815–840. https://doi.org/10. 1111/j.1468-5957.2010.02196.x
- Simunic, D. A. (1980). The pricing of audit services, theory and evidence. *Journal of Accounting Research*, 18(1), 161–190. https://doi.org/10.2307/2490397

- Simunic, D. A., & Stein, M. T. (1996). The impact of litigation risk on audit pricing: A review of the economics and the evidence. Auditing: A Journal of Practice and Theory, 15, 119–134.
- Stanley, J. D. (2011). Is the audit fee disclosure a leading indicator of clients business risk? Auditing: A Journal of Practice and Theory, 30(3), 157–179. https://doi. org/10.2308/ajpt-10049
- Tabachnick, B. G., & Fidell, L. S. (2001). Using multivariate statistics. Allyn and Bacon.
- Tee, C. M., Gul, F. A., Foo, Y.-B., & Teh, C. G. (2017). Institutional monitoring, political connections and audit fees: Evidence from Malaysian firms. International Journal of Auditing, 21(2), 164–176. https://doi.org/10.1111/ijau.12086
- Velury, U., & Jenkins, D. (2006). Institutional ownership and the quality of earnings. *Journal of Business Research*, 59(9), 1043–1051. https://doi.org/10.1016/ j.jbusres.2006.05.001
- Villalonga, B. (2019b). Ownership structure and corporate performance: Looking back and looking forward. Journal of Corporate Finance, 58, 64–67. https://doi. org/10.1016/j.jcorpfin.2019.04.005

- Warfield, T. D., Wild, J. J., & Wild, K. L. (1995). Managerial ownership accounting choices, and informativeness of earnings. *Journal of Accounting and Economics*, 20 (1), 61–91. https://doi.org/10.1016/0165-4101(94) 00393-J
- Whisenant, S., Sankaraguruswamy, S., & Raghunandan, K. (2003). Evidence on the joint determination of audit and non-audit fees. *Journal of Accounting and Research*, 41(4), 721–744. https://doi.org/10.1111/ 1475-679X.00121
- Xu, Y., Carson, E., Fargher, N., & Jiang, L. (2013). Responses by Australian auditors to the global financial crisis. Accounting and Finance, 53(1), 301–338. https://doi.org/10.1111/j.1467-629X.2011. 00459.x
- Yin, L. C., & Hung, L. W. (2011). Association between ownership structures, board independence and audit fees in Hong Kong SAR. Hong Kong Baptist University.
- Zhong, K., Gribbin, D. W., & Zheng, X. (2007). The effect of monitoring by outside blockholders on earnings management. Quarterly Journal of Business and Economics, 46(1), 37–60. https://www.jstor.org/ stable/40473429

🔆 cogent - oa

© 2023 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format. Adapt — remix, transform, and build upon the material for any purpose, even commercially.

The licensor cannot revoke these freedoms as long as you follow the license terms.

•

Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions

You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Cogent Economics & Finance (ISSN: 2332-2039) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- · Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com